Author’s response to reviews

Title: Efficacy of Removing Candida albicans from Orthodontic Acrylic Bases: An in vitro study

Authors:

Abdul Razzak Ghazal (dr-ghazal@hotmail.com)
Ghassan Idris (ghassan.idris@otago.ac.nz)
Mohammad Hajeer (myhajeer@gmail.com)
Karam Alawer (drkaramalawer@gmail.com)
Richard Cannon (richard.cannon@otago.ac.nz)

Version: 2 Date: 17 Aug 2018

Author’s response to reviews:

Ethics approval and consent to participate section of the Declarations was added as requested, clarifying the approval number and that the consent form was both verbal and written.